

SONDERFORSCHUNGSBEREICH 504

Rationalitätskonzepte, Entscheidungsverhalten und ökonomische Modellierung

No. 07-27

MIND THE GAP: THE EFFECTIVENESS OF INCENTIVES TO BOOST RETIREMENT SAVING IN EUROPE

Axel Börsch-Supan*

May 2004

Financial support from the Deutsche Forschungsgemeinschaft, SFB 504, at the University of Mannheim, is gratefully acknowledged.

*Sonderforschungsbereich 504, email: boersch-supan@mea.uni-mannheim.de



Universität Mannheim L 13,15 68131 Mannheim

MIND THE GAP: THE EFFECTIVENESS OF INCENTIVES TO BOOST RETIREMENT SAVING IN EUROPE

Axel Börsch-Supan

*Mannheim Research Institute for the Economics of Aging (MEA), Mannheim, Germany [#]National Bureau of Economic Research (NBER), Cambridge, Massachusetts, USA

ABSTRACT

Pension reforms all across Europe have a common theme: to reduce the generosity of the pay-as-yougo public pension pillar threatened by population aging, and to build up new pillars by private saving through occupational and individual pension plans. The extent of such retirement saving varies a great deal across Europe. This variation reflects, among other factors, the differences in public pension systems, taxation and capital market regulations. The first part of this paper looks at this variation in an attempt to learn about the effectiveness of the various incentives to boost retirement saving. While we find a strong correlation between the generosity of pay-as-you-go pensions and retirement saving, there is no straight correlation between the volume of retirement saving and the extent to which it is tax-favored. The second part of the paper uses the recent reforms in Germany as "experiments" that may shed light on which incentives might work and which might fail. We describe the introduction of the tax-favored "Riester pension plans" in 2001 and the 2004 tax reform, which changes the tax treatment of retirement savings in Germany from a conventional to a deferred taxation scheme. In spite of a deep subsidy and a generous tax treatment, "Riester pensions" have not found much attraction, while the originally heavily tax-favored whole life insurance is still wide spread. We conclude that boosting retirement saving requires more than simply tax relief.

ADDRESS:

Prof. Axel Börsch-Supan, Ph.D. MEA, L13,17 University of Mannheim 68131 Mannheim Fax: +49-621-181-1863 Email: axel@boersch-supan.de

ACKNOWLEDGEMENTS: This paper was written for the OECD workshop on tax-favored retirement saving plans, 28. May 2004. We are grateful for helpful comments by xxx and the participants of the OECD workshop. Financial support was provided by the National Institute on Aging (NIA) through the NBER, the German Science Foundation (DFG) through the Sonderforschungsbereich 504, the State of Baden-Württemberg and the German Insurers Association (GDV). The usual disclaimer applies.

Mind the Gap: The Effectiveness of Incentives to Boost Retirement Saving in Europe

by Axel Börsch-Supan

1. Introduction

Pension reforms all across Europe have a common theme: they reduce the generosity of the pay-as-you-go (PAYG) public pension pillar threatened by population aging, and they attempt to build up new pre-funded pillars that rely on private saving through occupational and individual pension plans. The extent of such retirement saving in Europe is the topic of this paper. It varies a great deal across Europe. This variation reflects, among other factors, large differences in public policy, notably the generosity of public pension systems, capital taxation and capital market regulations. Can we learn from these differences? Do they tell us about the effectiveness of different policy approaches in order to boost retirement saving?

How to boost retirement saving is an important question. Generating a significant increase in retirement saving is necessary to compensate for the cuts in PAYG pensions which in turn are needed to keep future contribution rates bearable. We will see that the amounts necessary are large. However, saving now cuts into consumption now, so people tend to procrastinate, in particular, because the force of compound interest – and thus the power of saving for retirement early on – is hard to intuitively understand. Which incentives are needed to overcome this tendency to procrastinate and behave myopically? This is both a normative and a positive question. The normative aspect deals with the question which retirement income level the state should enforce. Almost all in Europe will agree that society has the obligation to provide a subsistence income. This, however, is not the point here because – even after the current string of pension reforms have taken their bite – most European countries will still provide levels of mandatory PAYG pensions that are well above subsistence levels. Should the state enforce private saving to guarantee retirement incomes substantially above subsistence levels?

There are a number of reasons why one might give preferential tax treatment to retirement savings. The first argument applies to savings formation in general. The savings rate is very low in many countries.¹ An inadequately low savings rate leads to a high level of individual

¹ The USA provides the best known example.

consumption in the short run, but also leads to a reduction in investments which are required for the high levels of long-term economic growth which pave the way for consumption at a later date. It is difficult to define in practice how high the ideal savings rate should be. As long as the rate of return on capital continues to be higher than the growth in total wage income we can be sure, however, that this ideal savings rate has not yet been reached. This is the case in almost all the OECD countries, including in the Federal Republic of Germany (Abel, Mankiw, Summers and Zeckhauser, 1989). Moreover, in order to achieve neutrality between consumption today and later as well as neutrality between consumption of present and future generations, deferred taxation of savings is required.

The second argument is that many people are myopic and only begin to make provision for their old age when it is too late. Even if the paternalism implicit in this view might be unappealing for some, there is plenty of empirical evidence corroborating myopia. A more formal argument is that tax relief will mitigate the negative effects of liquidity constraints. An additional factor is that most people have little intuitive grasp of interest and compound interest mechanisms and believe it is possible to compensate for the low contributions made during the first half of their working life by saving twice as much during the second half, which it is not. As it is not possible for the individual citizen to rectify this mistake once it has been made, paternalistic logic suggests that the state must encourage saving.

Third, **i** also may make sense from a purely fiscal standpoint to deploy tax instruments to ensure adequate old-age provision. If retirement income drops below a defined threshold, the law in most European countries currently requires that inadequate incomes be topped up by social assistance payments. This should only be an emergency measure, however, as otherwise the far-sighted will simply end up subsidizing the short-sighted. In federal systems like Germany this would also result in fiscal side effects shifting burden from the federal to the state and local level because social assistance is financed from states and local authority resources.

Finally, there is the more subtle argument of adverse selection. It is based on the observation that people who believe they have an above-average life expectancy are more likely to convert funded old-age pension provision into life annuities. Others prefer lump-sum payments. This means that private life annuities are too expensive for most people.² This adverse selection of "poor risks" can be rectified by giving preferential tax treatment to life annuities rela-

² E.g. Walliser und Winter (1999).

tive to lump-sum payments, assuming we wish to avoid the instrument of compulsion and the negative incentive effects associated with it.

In those countries in which pay-as-you-go systems are gradually being supplemented by funded systems, tax allowances for old-age pension provision have also been used to accelerate the transition from one system to the other and to spread the transition burden equally across several generations.³ Whether these tax concessions are self-financing is a matter of controversy, see Boskin (2003) versus Auerbach, Gale and Orszag (2003).

This paper focuses on the positive aspects, given that the normative question has been answered affirmatively. Are tax-incentives sufficient to boost retirement saving? What are the experiences in Europe with different tax regimes, such as conventional and deferred taxation? Which other instruments have been used, such as mandatory, quasi-mandatory, opt-in and opt-out schemes? Are such instruments necessary at all, or will people adjust their retirement savings automatically to the new environment of lower public pensions? What is the role of information about the pension system in the willingness to accumulate own retirement savings?

This paper thus takes a fairly broad view on incentives to boost retirement saving not **e**-stricted to tax privileges. The paper is structured as follows. After documenting the savings volume necessary to compensate for the cuts in pay-as-you-go pensions in section 2, the first part of this paper uses the cross-national variation in Europe to address these issues. Section 3 compares the current tax and pension regimes and relates them to the size of the "three pillars", i.e., public, occupational and private pensions. Section 4 looks at the substitution between PAYG pensions and private retirement savings, section 5 at the substitution between retirement savings and other savings. Section 6 discusses saving incentives other than taxes.

The second part of the paper tries to learn from the recent reforms in Germany. They are a rather incisive change in pension regime from a monolithic PAYG system to a true multipillar system. It will eventually yield a valuable basis for program evaluation. The reform process included the introduction of private retirement accounts ("Riester pensions") in 2001 and the 2004 tax reform, which changes the tax treatment of retirement savings in Germany from a conventional to a deferred taxation scheme. Section 7 describes the status of retirement saving before the 2001 reform, section 8 describes the 2001 and 2004 reforms, and section 9 attempts

³ Examples may be found in the United Kingdom and Hungary (Palacios and Rocha, 1998).

a preliminary assessment of the reforms, obviously at a very early stage. Section 10 summarizes the main lessons and concludes the paper.

2. How much retirement saving is needed to stabilize the public pension systems?

This section briefly sets the stage for parts 1 and 2 of this paper by computing an important benchmark number: How much saving for old age is necessary to stabilize the ailing PAYG systems in Europe? These systems face a common problem. Due to population aging, the system dependency ratios will sharply increase, often about double. Since solutions within the PAYG systems are limited due to sheer size of the dependency increase, almost all pension reform attempts in Europe make use of the added flexibility gained by pre-funding.⁴

We use Germany as an example for a numerical calculation of the volume of pre-funding necessary to keep the PAYG contribution rate flat during the aging process. Such computations obviously depend on the policy mix between parametric and fundamental reform steps.⁵ We employ a partial equilibrium analysis using a policy mix proposed by Birg and Börsch-Supan (1999) which comes close to the current reform path in Germany. Within the PAYG system, the effective retirement age will be increased by 3 years until 2035. The PAYG contribution rate will be frozen from 2005 on. This will reduce the replacement rate in the PAYG system by about 20% until 2035. Together with the existing second and third pillar pensions, this will create a mix in Germany similar to what it is now in the Netherlands and Switzerland.

The savings volume necessary to fill this pension gap is depicted in figure 1 and depends on birth cohort. It is based on a financial instrument similar to a group life insurance which covers all three biometric risks (longevity, disability and survivorship) and is paid out on retirement as an annuity. For the sake of simplicity our computations are based on a standardized life course, in which gainful employment begins at age twenty and ends at age 60, the latter

⁴ Privatization and pre-funding are often used to describe the move toward funded systems. However, they are associated with two different aspects: privatization is the creation of funded individual accounts, pre-funding means closing the gap between social security benefits promised to date and the assets on hand to pay for them. One can easily imagine combinations of privatization and pre-funding during the transition to a partially funded system, ands this is exactly what is simulated in the sequel.

⁵ Parametric reform steps within the PA YG system, in particularly an increase in the retirement age, can go a long way in stabilizing future contribution rates. Unlike to the UK, however, they cannot absorb all the burden in the countries most affected by population ageing. Germany, for example, would need a shift of the average retirement age by 9.5 years in order to fully compensate the increase in the dependency ratio, see Börsch-Supan (2000).

gradually increasing to age 63 in the year 2035. We apply the upper variant of the life expectancy projections used by the most recent government computations. Hence, this insurance covers the individual risk of longevity as well as this projected aggregate increase of longevity. Early disability occurs between age 45 and 60 with an increasing probably averaging 15 percent, the current frequency. Survivor benefits are paid in accordance to the average current probabilities which is likely to be an overestimate because an increasing share of women will have their own pensions.⁶



Figure 1: Saving rates necessary to fill pension gap (Percent of gross earnings, by real rate of return)

The pension or insurance company is investing the accumulated capital in the market using a broad portfolio of stocks, bonds, direct placement and real estate. We assume that this portfolio generates the average gross rate of return that prevailed between 1980 and 1995 in the industry. This was 6.5% in real terms – however, we will also vary this rate in our projections. We subtract administrative costs of 6% of contributions, the average for group insurance policies, ⁷ resulting in a net rate of return to the customer of 4.5% in real terms. One may find this overly optimistic in these days, hence, we also present a simulation with a slightly higher (5.5%) and a substantially lower real rate of return (3.0%). We also assume that this rate will

Source: Birg and Börsch-Supan (1999).

⁶ In this case, Germany has a partial offset rule which reduces total claims.

⁷ Administrating individual policies is more expensive. This is why we choose group policies that can be bundled by employers, as is done in the Netherlands or, in a slightly modified form, through the US 401(k) plans.

decline by 0.5 percentage points in the wake of population ageing. This estimate is based on the growth model by Börsch-Supan, Ludwig and Winter (2003).

Figure 1 shows that the earlier cohorts need to fill a smaller gap, while the later cohorts have longer time to exploit the force of compound interest. The implied saving rate (i.e., contributions to new pension accounts, expressed as a percentage of gross earnings to be compatible to the PAYG contribution rates) is highest for the 1965 cohort and reaches about 3.2% of gross earnings if the past rate of return prevail, but exceed 4.6% of gross earnings if only a 3% real rate of return can be achieved. These two figures bracket the saving rate which the German government is proposing (4% of gross earnings).

We conclude that the volume in household saving to be generated by tax or other incentives is substantial in Germany. Extrapolating this simulation to France and most other European countries will generate slightly smaller gaps to be filled because aging is more pronounced than in the EU average. Aging will, however, be stronger in Italy where an even higher volume of saving needs to be generated in order to fill the pension gap.

Part I: Evidence on Tax-Favored Retirement Saving in Europe

The extent of tax-favored retirement saving varies a great deal across Europe. This variation reflects, among other factors, the differences in public pension systems, taxation and capital market regulations. The first part of this paper looks at this variation in an attempt to learn about the effectiveness of the various incentives to boost retirement saving.

3. Retirement saving and the institutional setting in Europe

This introductory section looks at the crude cross-national correlation between the extent of retirement saving, pension systems and taxation. We begin with the large variation in pension schemes. Table 1 shows the mixture between PAYG and funded old-age provision across seven selected European countries and the United States.

Percent^a D F Ι Ε NL GB CH First Pillar^b 65% ^f 79% 85% 74% 92% 50% 42% 6% ^e Second Pillar^c 5% 1% 4% 40% 32% 25%

25%

Table 1: Sources of Retirement Income – Size of the "Three Pillars"

15%

Third Pillar^d

10%

Notes: (a) Percent of total income of average two-person household just after retirment. (b) Public retirement income (public pensions, social assistance, civil servants' pensions, etc.). (c) Private occupational pension income. (d) All other retirement income (asset income, net transfers received, earnings, etc.). (e) In France, mandatory occupational pensions are pay-as-you-go financed and are included in the first pillar. (f) In Great Britain, first pillar income also includes SERPS. (g) In the US, 25 percentage points of this figure are earnings. *Sources*: Disney et al. (1998), Gruber and Wise (1999), Börsch-Supan and Miegel (2001).

4%

10%

26%

In most Continental European countries, notably in the three largest countries France, Germany and Italy, the PAYG mechanism is the most important instrument. There are notable exceptions, however, such as the Netherlands and Switzerland. In spite of these differences, table 2 shows an astounding similarity in the overall replacement rates. Adding up the income from all pillars, we find that households just after retirement receive approximately 80% of the income they enjoyed shortly before retirement. This replacement rate is nearly identical in Germany, France, Italy, the Netherlands and in Switzerland; it is slightly lower in Great Britain.

US

45%

13%

42%^g

10%

1				-				
	D	F	Ι	Ε	NL	СН	GB	US
Percent of Preretirement ^a	82%	79%	80%	./.	78%	81%	69%	./.

 Table 2: Comprehensive Retirement Income Replacement Rate

Notes: All income sources of average two-person retiree household just after retirement as percent of total income of average two-person household just before retirement. Source provides no strictly comparable data for Spain and US. *Source*: Disney et al. (1998).

The combination of tables 1 and 2 suggests a strong substitution among the three pillars, that is, a strong adaptation of the private individual and employer-based old-age provision to the public pension system.

Table 3: Assets and Participation in Tax-Favored Retirement Saving Plans

	D	F	Ι	Ε	NL	СН	GB	US
Percent of GDP	8.6%	5.7%	5.5%	5.0%	116%	112%	76%	108%
Percent of employment				42%	90%	79%	59%	53%

Source: Antolin, de Serres and de la Maisonneuve (2004). Data refers to year 2000. Life-insurance not included.

Tables 3 and 4 tell a similar story from a slightly different point of view. The accumulated assets in tax-favored retirement saving plans (table 3) are much smaller in those countries which have large public pension replacement rates (table 4).

 Table 4: After-Tax Replacement Rates (Public Pensions)

	D	F	Ι	Ε	NL	СН	GB	US
Percent	77%	77%	97%	92.5%	46%	57%	40%	48%

Source: Adapted from Casey (2003). Based on the salary of an average production worker at "normal" age of retirement. In France, mandatory occupational pensions are pay-as-you-go financed and are included in the above figure. In the Netherlands, mandatory occupational pensions are pre-funded and not included in the above figure.

Relative to the force of public pension replacement rates, taxation does appear to not play a major role in explaining the cross-national variation in the importance of tax-favored saving plans. Table 5 shows tax regimes and effective tax-rates, using the familiar "EET" notation of deferred taxation. Germany and France apply an even more lenient taxation than the five countries with deferred taxation, and Spain has lower tax rates than the two Anglo-Saxon

countries, nevertheless do play tax-favored savings only a minor role in Germany, France and Spain.

	•			-				
	D	F	Ι	Ε	NL	СН	GB	US
Percent	EEP	EEP	EPP	EET	EET	EET	EET	EET
Effective Tax on Contributions				22.1%	37.1%	16.6%	22.1%	29.0%
Effective Tax on Accrued Income				14.5%	12.8%	12.7%	20.1%	22.3%
Effective Tax on Benefits				17.1%	32.1%	11.6%	17.1%	24.0%

Table 5: Tax Treatment of Retirement Savings

Source: Yoo and de Serres (2004). E=tax exempt, P=partially exempt/partially taxed, T=taxed. Comparable data for Germany, France and Italy is not available.

We conclude from this introductory section, that while we find a strong correlation between the generosity of pay-as-you-go pensions and retirement saving, there is no straight correlation between the volume of retirement saving and the extent to which it is tax-favored. In the following, we give these correlations a closer look.

4. Substitution between PAYG and private pensions

Section 3 suggests that substitution between PAYG pensions ("first pillar") and private pensions (occupational and individual pensions in second and third pillar) is the most important retirement saving mechanism. This is in line with economic theory: models with consumption smoothing will predict substitution between PAYG and funded pensions, while the effect of taxation depends on the dominance of the substitution over the income effect.

One actually has to carefully distinguish two kinds of substitution or "crowding out", see the arrows in figure 2: (1) substitution within several types of "retirement wealth", including mainly notional pension wealth (claims on future PAYG pensions) and real wealth (claims on assets usable only for retirement, such as occupational and individual pensions), and (2) substitution between retirement wealth (defined by liquidity restrictions such as no availability before retirement age) and other wealth that can be liquidized freely.⁸ Both substitution

⁸ Bernheim (1987a and b) addressed a similar distinction, focusing on bequeathable and other wealth.

mechanisms are important in the pension reform discussion. The first substitution mechanism (people accumulate more retirement assets if PAYG pensions are reduced) is well visible in economic models with consumption smoothing, such as the well-known strand of overlapping generation models which include both pay-as-you-go pension schemes and life-cycle motive for retirement saving (Auerbach and Kotlikoff (1987) for the United States; Miles (1999) for Great Britain; Börsch-Supan, Ludwig and Winter (2003) for Germany). The second substitution mechanism (for example, people may buy smaller homes and thus accumulate less assets for a smaller down payment when they have to accumulate more assets for retirement) is not visible in these models since they have only one savings motive (consumption smoothing over the life-cycle) and do not distinguish between high- and low-frequency savings or "mental accounts" for different savings purposes.



Figure 2: Substitution among savings types ("Crowding out")

The combination of the two effects – arrow (3) in figure 2 – is the center of the crowding out debate between Feldstein (1974) and Barro (1974). Crowding out is an important policy issue. A central argument put forward in pension reform discussions is the macroeconomic superiority of a funded system vis-à-vis a PAYG system.⁹ This argument is valid only to the extent

⁹ Welfare superiority *including transition costs* requires additional effects over and above this capital accumulation, strengthening the necessity of *new* saving. See Fenge (1997), Börsch-Supan (1999) and Sinn (2000) for an exchange on this point.

that new savings is created which adds to the capital stock and therefore promotes economic growth. In this case, the generation of retirement savings through the first substitution mechanism is not offset by an equally large substitution away from other savings through the second mechanism. Moreover, if the contrary is true ("full crowding out") and pre-funding is incensed by tax relief, this is wasted because it only shifts existing savings from one form to another.

The cross-national evidence on the first substitution mechanism appears fairly solid as we have seen in tables 2 through 4. Further evidence comes from an inspection of life-cycle saving profiles.



Figure 3: Age-specific saving rates (cohort corrected)

Sources: France: Fall, Loisy, and Talon (2001); Germany: Börsch-Supan, Reil-Held, Rodepeter, Schnabel, and Winter (2001); Italy: Brugiavini and Padula (2001); Netherlands: Alessie and Kapteyn (2001).

Figure 3 shows, by age group, median saving rates in France, Germany, Italy and the Netherlands. They are based on a comparable longitudinal framework, represent life-cycle saving purged from cohort effects, and employ comparable variable definitions and data sources as part of the International Saving Comparisons Project (Börsch-Supan, 2001 and 2003). The saving profiles in France, Germany and Italy are rather flat and show no dissaving in old age. One possible explanation is that the high replacement rates of the public pension systems in these countries have made private retirement income largely unnecessary. If other saving motives, such as precaution and intergenerational transfers, are more important than retirement saving, age-saving profiles are likely to be much flatter than under the textbook life-cycle hypotheses which predicts saving in young and dissaving in old age. This explanation is in line with the work by Jappelli and Modigliani (1998) who argue that the main mechanism for "retirement saving" in Italy is the PAYG system. While we lack the most appropriate counterfactual – French, German and Italian data from times when these countries had no PAYG systems – figure 3 depicts the case of the Netherlands which have, as opposed to France, Germany and Italy, only a small base pension provided by their PAYG public pension system. All additional retirement income in the Netherlands has to be provided by (mandatory) savings plans, commonly provided through occupational pension plans. Figure 3 shows that the median Dutch household has a much more pronounced hump-shaped life-cycle savings profile than the median French, German and Italian households, and it exhibits dissaving among the elderly as they draw down their mandatory saving accounts.

This cross-national evidence suggests that a pension reform towards a multi-pillar system with a substantial portion of funded retirement income will revive the retirement motive for saving in France, Germany and Italy. In fact, as pointed out in section 2, these systems will look very similar to the current Dutch system. Hence, it is likely that saving rates among the young will increase (to accumulate retirement savings), and saving rates among the elderly will decline sharply (because they will dissolve their retirement savings). Combining the results of figures 1 and 3 gives us an order of magnitude for those effects. The net saving rate of the average German household in mid age would increase by about 4 percentage points from 11 to 15 percent, while the saving rate would decrease by about 6 percentage points from 4 to -2 percent in old age, when retirement accounts are drawn down for consumption.

From a policy point, there seems no doubt that people adjust their behavior to the generosity of public pension systems like in a system of communicating pipes. Taxes appear to be uncorrelated with these adjustments. The cross-national evidence, however, reflects long-run adjustments. Hence, this section does not necessarily imply that tax incentives boosting retirement savings are superfluous. While the evidence of table 5 - high taxes on retirement savings in the Netherlands and the United States – seems to be contradictory to the high levels of retirement saving, these tax rates have been established after a multipillar system had been installed.

5. Substitution between retirement saving and other saving

There is also clear evidence on the partial effects of tax incentives (see arrows (3), (4a) and (4b) in figure 2). Several econometric studies using large micro data sets have been shown the impact of tax relief on dedicated savings in Germany, e.g. whole life insurance policies and savings and loan agreements (Börsch-Supan and Stahl, 1991b; Brunsbach and Lang,1998; Lang, 1998; Walliser and Winter, 1999). These studies show quite clearly that tax relief for specific investment vehicles strengthen the attractiveness of the relevant forms of saving. Indeed, they appear to have markedly shaped the retirement savings landscape in Germany, see section 7.

This is, however, not the central issue. Evidence on the total effect – the generation of new savings, when retirement savings through the first substitution mechanism is not offset by an equally large substitution away from other savings through the second substitution mechanism, i.e., arrow (2) in figure 2– is still controversial, and a European perspective¹⁰ brings, rather unfortunately, not much additional light into the old controversy between Venti and Wise (1990) on one side, and Gale and Scholz (1994) and Attanasio and De Leire (1994) on the other side, summarized by Skinner and Hubbard (1996) for the US and by Attanasio and Banks (1998) for a comparison between the US and the UK.

Auerbach and Kotlikoff (1983) pioneered the discussion on the difficulties with the identification of the substitution parameters both in time series and cross sectional studies for the US. The structural instability of the parameters is due to the changes in population structure and to the fluctuations in interest rates as well as in contribution rates during the transition after the introduction of a new pension system.

Additional complications along these lines stem from the fact that a considerable share of elderly households in Continental Europe seem to be "overannuitized". Evidence by Börsch-Supan and Stahl (1991a) and Börsch-Supan (1992, 1994) shows that the annuity income of a majority of pensioners (mostly PAYG income provided through the public system) by far exceeds consumption expenditures. The next generation of pensioners appears to be aware of this fact, since saving rates in Germany drop for the now middle aged cohorts. Given this trend, a reduction in public pension wealth may not necessarily be channeled into new savings, or only after a transition period of new costly learning. At the same time, these house-

¹⁰ The following literature survey draws on Börsch-Supan and Brugiavini (2001).

holds, if left to their own devices, may also shun pension assets because they do not want to lock their wealth into long term saving for retirement. In short: a careful analysis of the link between total saving rates and changes in pension systems needs to be cohort-specific and cannot simply use aggregate time-series.

It may therefore not be a major surprise, if for Europe the empirical evidence is scanty and there exists a wide variety of results, making it particularly hard to reach any conclusion. Kim (1992) links changes in the retirement system to the savings rate and shows that the German pay-as-you-go system has crowded out saving to a significant extent. Cigno and Rosati (1996) confirm these findings but explain the crowding-out effect unconventionally by repercussions on fertility rather than through the familiar channels stressed by Feldstein (1974). Other time series studies include Rossi and Visco (1994) for Italy, who are able to attribute part of the decline occurred during the 1980s in the saving rate to the increased generosity of the social security in the 1970s. In other countries the relationship between pension wealth and private fungible wealth emerging from time series data is poorly determined.¹¹ Cross sectional analysis based on Italian micro data (Brugiavini, 1987; Jappelli, 1995) obtain low estimates of the substitution parameter. Recent studies for the Netherlands obtain mixed results. In particular, by using the Dutch Socio Economic Panel (SEP) data on private wealth and on (constructed) data on social security wealth and pension wealth, Alessie, Kapteyn and Klijn (1997) and Kapteyn, Alessie and Lusardi (1999) have investigated the question whether or not there exists displacement between discretionary private wealth on the one hand and social security and pension wealth on the other hand. For pension wealth they do not find any evidence of displacement, while for social security wealth they find full displacement.¹²

Clear cut "experiments" in which the pension system changes but everything else remains constant, are hard to come by also in Europe. Attanasio and Brugiavini (1997) rely directly on a test of how changes in saving rates are related to changes in pension wealth following the 1992 Reform of the Italian Social Security System in a "quasi-natural-experiment" setting. They find that households whose public pension wealth was substantially curtailed by the reform show a marked increase in their private saving rate. Germany will provide an "experiment" with its 2001 and 2004 reforms, but, as we will see in part 2 of this paper, it is too early to draw firm conclusions yet. So far, the German experience with its very small uptake rates

¹¹ See Kohl and O'Brien (1998) for a detailed survey.

¹² See also Euwals (2000). The evidence based on cross sectional data for the USA (Hubbard, 1986) and for Canada (King and Dicks-Mireaux, 1982), is also rather mixed.

of the new tax-favored retirement saving instruments shows that short-run effects maybe quite different from the long-run effects shown in the previous section.

Börsch-Supan and Lusardi (2003) attempt a cross-national analysis exploiting the cohortcorrected savings data in the International Savings Comparisons Project. They find that cohort-corrected saving rates are positively (!) correlated to public pension replacement rates, suggesting crowding in rather than crowding out – this positive correlation, however, changes sign to a significant negative one, once credit restrictions are taken account of.¹³

In summary, the European evidence so far does not add much to the evidence gathered in the US debate. This is in spite of the large institutional variation in Europe. While we do know that subsidies strongly increase saving in the specific form that is subsidized, possibly to the detriment of other saving forms, we do not really have firm evidence that saving-related tax relief or similar subsidies increase total saving in Europe.

This does not make tax-relief a potentially wasteful instrument. Even if tax relief would only shift other saving to retirement saving, this may be a valuable mechanism if the government wants to make sure that the elderly will have a generous multipillar retirement income. That is, even if the government does not believe in the creation of new saving for macroeconomic purposes, it still may want to repress procrastination not only in form of consumption now, but also in the form of a larger house in the near future, and subsidize retirement consumption in the far future.

6. Other incentives: Opting in and out, mandatory savings

The U.S. experience, summarised by Wise (2001), has also taught us two further lessons. First, tax relief is not everything. The great success of individual retirement accounts (so-called "IRAs") and employer-sponsored retirement saving plans (so-called "401(k) plans") also seems to be due to information and advertising, and a consistent capital market regulation that reduced the uncertainty of investors. This is in great contrast to the UK experience, summarised by Disney (1996), where the lack of regulation and information led to the so-called mis-selling scandals that undermined investment in private accounts. In this scandal, inappropriate financial products were sold to households, often by door-to-door salespersons, resulting in huge financial losses to many families. State intervention seems necessary to provide

¹³ Specified as downpayment restrictions according to Chiuri and Jappelli (2000). The results should be cautiously interpreted since they rely on a small panel of six countries and five waves.

for a smooth working of the capital market. The German experience after the 2001 reform, to which we will turn in the second part of this paper, teaches a related lesson: lack of transparency – ironically created by overregulation – appears to have destroyed all positive incentives created by tax relief.

The second lesson from US experience is that tax relief only works for the upper two third of the income distribution. Neither IRAs nor 401(k) plans will have a noticeable impact on future retirement income of the lower third of the income distribution. In Germany and in Italy, these observations have sparked an extended discussion on whether to make private provision mandatory. The argument pro-mandatory savings is mainly the need to cover all persons in order to avoid pockets of poverty and to prevent moral hazard. The arguments against mandatory savings rest on the well known economic efficiency arguments against taxation.¹⁴ One advantage of private savings for old age as opposed to PAYG contributions is that the latter have mainly tax character (Börsch-Supan and Reil-Held, 2001), therefore create dead weight losses, while private saving does not. Making saving mandatory, however, introduces an element of force, thus shifts savings understood as voluntary insurance premia towards mandatory taxes.

The issue is complicated by moral hazard. As long as the government provides pension income of last resort, voluntary savings is a waste – from the individual point of view, at least among those in the lowest parts of the earnings distribution – if one can just as well apply for social assistance. People may therefore opt for mandatory savings in order to reduce moral hazard.

While most economists do not like the incentives costs associated with mandating retirement saving, Boeri, Börsch-Supan and Tabellini (2001 and 2002a) actually find that most workers in France, Germany, Italy and Spain prefer a pension reform with mandatory savings over a reform with voluntary savings. Besides the moral hazard argument, there are many other explanations for this finding, such as the already mentioned lack of self-control and the fear of procrastination, the expected higher likelihood of government bail-out if a pension fund fails in which mandatory savings were invested, etc. All of these arguments underscore a need for government intervention – most strongly, in the imposition of a mandatory saving plan; less strongly, by giving tax relief.

¹⁴ Summers (1989).

Part II: A Case Study of the Recent German Pension Reform

In this second part, we describe and analyze the 2001 and 2004 pension reforms in Germany. They are a good example for a rather incisive change in pension regime from a monolithic pay-as-you-go system to – eventually – a true multipillar system. Section 7 describes the status of retirement saving before the 2001 reform, section 8 describes the 2001 and 2004 reforms, and section 9 attempts a preliminary assessment.

7. Retirement saving before the 2001 pension reform

Tax-favored retirement saving in Germany takes several well-defined forms as occupational and individual pension plans. First, there are four types of occupational pensions: (I) direct pension promises based on company book reserves, (II) provident funds, independent organizations sponsored by employers, (III) direct insurance, where employers buy whole life insurance for their employees, and (IV) staff pension insurance, where employers set up their own insurance-like organization. All these plans are of the "defined benefit" type. Until 2001, defined contribution pension funds – an investment vehicle which is widely used in other countries – were not permitted in Germany. Table 6 gives an overview of their features.

Second, the only individual retirement saving form which was tax-favored before the 2001 pension reform is whole life insurance. Both accrual and benefits are tax exempt; moreover, contributions can be deducted from taxes up to a limit which depends on the tax-payers other dedicated saving instruments. Tax-favored and subsidized individual retirement accounts were introduced in the 2001 reform, see section 8.

Features	Investment Vehicles				
	Ι	II	III	IV	V
	Direct pension promise (Direkt- zusage)	Provident funds (Unter- stützungs- kasse)	Direct insurance (Direkt- versicherung)	Staff pension insurance (Pensions - kasse)	Pension funds (Pension- fonds)
Tax on con- tributions	Tax free		 Flat-rate tax Fully taxed but Riester subsidy/ tax deductable expense 	 Flat-rate tax Fully taxed but Riester subsidy/ tax deductable expense Tax free until 4% of BMG 	 Fully taxed but Riester subsidy/ tax deductable expense Tax free until 4% of BMG
Tax on benefits	Fully taxed		 Tax on re- turns only Fully taxed 	 Tax on re- turns only Fully taxed Fully taxed 	 Fully taxed Fully taxed
Investment	Inte	ernal		External	
Investment rules	None		Acc. Insurance Supervisory Act		None
Insolvency scheme	Membershij insurance f	p in pension fund (PSV)	Ň	lo	Membership in PSV
State super- vision	No	one	Federal Insu sichtsamt fi	rance Authority ir das Versicheru	(Bundesauf- ungswesen).

 Table 6: Types of occupational pension systems

Note: BMG = earnings threshold. *Source:* Börsch-Supan and Wilke (2003).

The institutional breadth of retirement savings and the various forms and levels of taxation imposed on them have significant monetary effects. These are calculated in Table 7 and expressed as after-tax benefits in percent of contributions accumulated over the average length of a work life. For ease of comparison, the calculation assumes an identical rate of return of 3% (including the pay-as-you-go scheme).

	Public retirement insurance	Civil ser- vant pe n- sion	Occupational pension I/II (pension promise/ provident fund)	Occupational pension III/IV (direct insurance/ staff pension insurance)	Whole life insurance	Investment fund
Tax Regime	EEP	EET	EET	TET	PEE	TTE
Benefits as percent of contributions	173.3%	143.7%	159.9%	142.8%	143.7%	124.8%

Table 7: Effects of different taxation rules on retirement saving

Note: The figures reflect a savings period of 37 years and an annual contribution of EUR 500, a real rate of interest of 3%, a retirement age of 65 and an average tax rate of 22%. Deductions are calculated on the basis of statutory percentage rates; no account is taken of maximum tax allowance amounts. E=tax exempt, P=partially exempt/partially taxed, T=taxed.

Source: Computed from Börsch-Supan and Lührmann (2000)

Table 7 shows that the current situation does not resemble a "level playing field". Public retirement insurance receives the most preferential tax treatment; investment funds (including pension funds) come off worst. The difference in the net benefit payments from the public retirement insurance scheme and an investment fund is almost 50% of contributions. The net benefit payments of pension forms which are subject to deferred taxation – public retirement insurance, civil servants pensions, and occupational pensions – tend to be higher than net benefit payments subject to the other tax principles. Public retirement and occupational pensions generate the highest net benefit payments as only a very minor share of the pension benefits (a hypothetical interest portion) is taxed and in many cases, thanks to generous exemptions, no tax is levied at all.

The unequal tax treatment of different forms of old-age pension provision cast occupational pensions and investments in investment funds in a highly unattractive light. While the unequal tax treatment of a pure capital investment in equities might be justified because this form of saving does not cover biometric risks, the other two investment vehicles do in fact cover these risks.

The asymmetric taxation of state and private old-age pension provision is not only reflected in different levels of net benefit payments but also exercises incentive effects and triggers substitution effects. These are apparent from the spread of different forms of private old-age pension provision in Germany. Of those investing in private forms of pension provision, 71.2%

have a life insurance policy but only 15.1% an equity or other investment fund. Occupational pensions are the weakest pillar of pension provision in Germany. The volume of such schemes is less than half of that of life insurance policies and less than 10 percent of public retirement insurance expenditure.¹⁵ Among them, the lion share of more than two thirds goes to type-I defined benefit pension plans (direct pension promise backed by book reserves) which also provide the highest tax advantage, see table 7.

The composition of financial wealth of German households excluding occupational pension wealth is displayed in Table 8. The most important component is whole life insurance, about a third of gross financial wealth. This is in line with the econometric studies by Brunsbach and Lang (1998) and Walliser and Winter (1999) who exploit cross-sectional variation in tax rates and find a significant influence of the favorable tax treatment on the portfolio share of whole life insurance in German households. Bonds make up the lions' share in this category, while stocks are less than 10 percent of the average household portfolio. This fact is also significant for financial markets, as life-insurance companies have not been allowed to invest significantly in stocks in the past, which in turn is one of the main reasons for thin capital markets in Germany. Stocks and bonds are tax privileged in so far as capital gains are tax exempt if the underlying asset has been held for longer than one year.¹⁶

,	• •			
1978	1983	1988	1993	1998
33,8%	26,7%	25,2%	17.5%	22,2%
13,6%	13,0%	9,5%	7.5%	7,5%
16,2%	19,6%	19,7%	31.4%	24,3%
36,4%	36,7%	42,4%	33.3%	30,7%
0,0%	4,0%	3,4%	10.4%	15,3%
100.0%	100.0%	100.0%	100.0%	100.0%
	1978 33,8% 13,6% 16,2% 36,4% 0,0% 100.0%	1978 1983 33,8% 26,7% 13,6% 13,0% 16,2% 19,6% 36,4% 36,7% 0,0% 4,0% 100.0% 100.0%	1978 1983 1988 33,8% 26,7% 25,2% 13,6% 13,0% 9,5% 16,2% 19,6% 19,7% 36,4% 36,7% 42,4% 0,0% 4,0% 3,4% 100.0% 100.0% 100.0%	1978 1983 1988 1993 33,8% 26,7% 25,2% 17.5% 13,6% 13,0% 9,5% 7.5% 16,2% 19,6% 19,7% 31.4% 36,4% 36,7% 42,4% 33.3% 0,0% 4,0% 3,4% 10.4% 100.0% 100.0% 100.0% 100.0%

Table 8: Composition of household wealth, Germany, 1978–1998

Note: Household data from the Einkommens- and Verbrauchsstichprobe (EVS).

Source: Börsch-Supan, Reil-Held, Rodepeter, Schnabel and Winter (1999) and own computations.

¹⁵ DIA (1999).

¹⁶ This has recently been changed to two years.

8. The German reforms in 2001 and 2004¹⁷

The core of the recent reforms includes two elements. The first core element are significant benefit cuts in the PAYG pillar. The replacement rate is successively reduced, implying benefit cuts of about 20% by the year 2040, the projected peak of population aging in Germany. This is achieved by a change in the benefits indexation formula in two stages, 2001 with the introduction of the "Riester steps" and 2004 with the introduction of the "sustainability factor", see Börsch-Supan and Wilke (2003).

The second core element of the Riester reform is the introduction and significant promotion of supplementary funded private pensions on the basis of individual retirement accounts ("Riester pensions") to fill the pension gap created by the benefit cuts. The objective is to offer incentives for people to take out supplementary private pension cover which, in the long term, should compensate for the future cuts in public pensions. These pensions are voluntary: there will be no legal mandate for people to invest in these additional private schemes. The Riester pensions resemble IRAs in the U.S.

The tax-relief and/or subsidies for Riester pensions can also be used for several types of occupational pensions (direct insurance, staff pension insurance and pension funds, see table 6), which merges elements of the existing German DB plans with elements of US-style 401(k) plans.

8.1 Individual retirement accounts ("Riester pensions")

The Riester pensions are heavily regulated and many restrictions apply. The most incisive restriction is on payment plans. Since additional private pension schemes are intended to supplement or replace benefits from the public pension scheme, the government decided that incentives will only be available for investment vehicles which guarantee payment of a life annuity payable from the date of retirement. Investment vehicles which provide for lump-sum disbursements are not subject to state subsidies.¹⁸ This restriction has already met with considerable criticism in the public debate as it excludes other forms of provision for old age (such as investments in old-age or nursing homes).

¹⁷ This and the following section draw on Börsch-Supan and Wilke (2003) who provide further details.

¹⁸ If a lump-sum payment is chosen, all subsidies have to be reimbursed to the tax authorities.

The incentives provided by the state can take two forms: direct savings subsidies or taxdeductible special allowances. The tax authorities automatically compute which of the two forms versions is most advantageous.

Direct savings subsidy. All dependently employed and certain self employed workers who pay personal contributions to a certified retirement pension policy are entitled to receive a direct retirement savings subsidy. The subsidy is paid directly into the beneficiary's saving account. A basic subsidy and a child subsidy for each child for which child benefits were received during the previous year is paid. Child subsidies are payable to the mother. In the case of married couples, both partners receive a basic subsidy if they have each taken out their own supplementary private pension policy. In addition, non-entitled partners (such as mothers not in paid employment) are also entitled to receive the full subsidy for their own retirement pension policy provided that the respective married partner subject to compulsory insurance contributions has paid his or her minimum personal contribution to their supplementary retirement pension policy (see below).

From on	Savings rate	Basic subsidy [Euro p.a.]	Child subsidy [Euro p.a.]
2002	1 percent	38	46
2004	2 percent	76	92
2006	3 percent	114	138
2008	4 percent	154	185

m 11	•	D ¹	•	1 • 1•
Tahlo	U٠	Inroct	cawinac	C11hC1d10C
Iuvie	1.	Dueu	surings	substates

Source: Börsch-Supan and Wilke (2003).

Table 9 shows the maximum incentive subsidies available as of 2002. In order to qualify for the maximum subsidy the beneficiary must invest a specified percentage of his or her gross earnings (denoted as "saving rate"). This percentage increases until 2008 in four steps. The percentage is applied to the actual earnings level, capped at the same cap as the PAYG contributions are (about 2 times average earnings). If less money is invested, the state subsidy is reduced accordingly. The scheme is complicated by the fact, that the subsidy is included in the savings amount. Hence, the actual saving rate necessary for the maximum subsidy is lower than the percentages indicated in the second column of table 9. In turn, certain minimum amounts are necessary, see table 10.

Year	No child	One child	Two or more children
2002 - 2004	45	38	30
As of 2005	90	75	60
Courses Dönsch Curson on	$4 W(11_{ro} (2002))$		

Table 10: Minimum Savings

Source: Börsch-Supan and Wilke (2003).

Tax deductible special expenses. Alternatively, qualifying retirement savings can be deducted as "special allowances" from income taxes. This is usually more advantageous for workers with higher than average earnings. Saving rates, caps etc. are the same as in the subsidy case. Table 11 shows the maximum tax-deductible contributions to private retirement savings accounts.

Table 11: Maximum Savings

From on	Tax deductible special expenses in Euro/Year	
2002	525	
2004	1.050	
2006	1.575	
2008	2.100	

Source: Börsch-Supan and Wilke (2003).

Criteria for individual pension plans eligible for subsidies/tax relief. Individual retirement accounts only qualify for state promotion if they meet criteria laid down in the new Certification of Retirement Pension Contracts Act ("AltZertG"). It contains a long list of rules which make the system complex for customers and potential insurers alike. Qualifying pension plans require certification by the Federal Financial Markets Authority ("Bundesanstalt für Finanzdienstleistungs- und Finanzmarktaufsicht") which will be granted automatically if they fulfill the following preconditions:

- 1. The investor must be committed to making regular, voluntary pension contributions.
- 2. Pension benefits may only be paid out when the beneficiary reaches the age of 60 at the earliest or upon reaching retirement age.
- 3. At the beginning of the disbursement phase, the accrued pension contributions (inclusive of subsidies) must be guaranteed (i.e., the nominal rate of return must be nonnegative).
- Pension payments must guarantee lifelong benefits which retain or increase their nominal value, i.e. in the form of a life annuity or disbursement plan linked to lifelong annual installments.

- 5. The disbursement plan must continue to provide benefits until the beneficiary reaches the age of 85 and subsequently provide a life annuity guaranteed by the capital available at the beginning of the disbursement phase.
- 6. Supplementary survivor's coverage must not have features which offset the original plan.
- 7. Initial commission and administrative charges must be spread equally over a period of at least 10 years.
- 8. The investor must be informed about the following issues before taking out the policy: The level and distribution over time of commission and administrative costs, the cost of switching to a different policy, the costs of financial management, the costs involved in changing to a different insurer.
- 9. The investor must be informed once a year during the term of the policy about how his or her contributions are being used, capital formation, costs and yields, and also about whether and to what extent the insurer takes account of ethical, social and ecological investment criteria.
- 10. The investor must have the right to suspend contributions during the saving phase, to allow the policy to continue running without making additional contributions, or to terminate the policy by serving three months notice to the end of the quarter.
- 11. Policy rights may not be assigned or transferred to third parties. Claims to pension benefits cannot, as a result, be bequeathed.

Products eligible for subsidy support and into which old-age pension contributions and the proceeds on such contributions may be invested include pension insurance and capitalization products, bank accounts with accumulated interest and shares in growth and distributing investment funds. These products are offered by life insurance companies, banks, capital investment companies, financial services institutions and securities services companies.

8.2 State promotion of occupational pension schemes

The Riester reform remained largely undecided on the role of occupational pensions versus individual accounts. Traditionally, occupational pensions have played a minor role in Germany, particularly in comparison with other countries. Demand for participation in occupational pension schemes has also been falling in recent years.¹⁹ On the other hand, occupational pensions may provide a psychological substitute for mandated private pensions. In order to strengthen occupational pensions, additional (implicit and explicit) subsidies were introduced with the Riester reform.

¹⁹ See Ruppert (2000).

The Riester reform introduced pension funds as a vehicle for occupational pensions – an investment vehicle which is widely used in other countries, but was not permitted in Germany. There are now five different investment vehicles in German occupational pension schemes, but only three are eligible for Riester incentives: direct insurance, staff pension insurance and pension funds (see table 6). As the employer has to provide the employee with the possibility to benefit from the Riester incentives, this means – especially for smaller companies – that some companies now have to restructure their pension schemes.

The most important change is the general right to convert part of the salary directly into contributions to pension plans. This applies regardless of whether the contributions are paid by the employer or the employee. Arrangements may be based both on gross or net pay. If they are based on net pay, there is a large implicit subsidy since the so-converted salary may not only be subject to deferred taxation but can also be exempt from social security contributions, at least until 2008. If they are based on gross pay, contributions may enjoy the same direct subsidies or tax relief as contributions to individual accounts, as long as the occupational pensions meet certain criteria which are less restrictive than the criteria for individual pension plans. Which contribution rules apply depends on the chosen investment vehicle and the incentives they attract (see table 6). Collective bargaining agreements, however, have precedence over the right to convert salary. This means that an employee covered by a binding collective agreement is only entitled to convert his or her pay into pension if this is explicitly provided for in the terms of the collective agreement. This rule makes sure that employers and unions can impose their own rules on occupational pension plans.

8.3 Deferred taxation

Contributions to the Riester pensions introduced in 2001 will be tax exempt during the saving phase, pension payments during the benefit phase will be taxed in full as normal income. This applies to all benefits regardless of whether these accrue from contributions, subsidies or capital gains. One may regard this as another form of subsidy, since taxes occur later in life (hence, an implicit tax credit) and usually at a lower rate due to progressivity.²⁰

The 2004 tax reform has introduced tax-deferred taxation to all other retirement saving vehicles as well, including public pensions. This requires a complex transition process. Currently, public pensions are taxed "EEP", see table 5, 6 and 7. Only a part of the hypothetically ac-

²⁰ Börsch-Supan and Lührmann (2000). The "tax credit" feature depends on the an income or consumption tax point of view.

crued interest is taxed after retirement (this part is assumed to be 27% of benefits if retirement takes place at the "normal" age of 65; 32% at early retirement with age 60). In 2005, this taxable fraction will jump to 50%, and successively increase to 80% in 2020 and 100% in 2040. During a substantially shorter time span, contributions to all retirement schemes (public, α -cupational and individual) will become tax deductible, see figure 4 It begins with 60% of contributions in 2005 and successively increases to 100% in 2025. This scheme avoids double taxation (with a few exceptions), but incurs substantial revenue losses (about 1 billion Euro per year until 2006).



Figure 4: Transition to deferred taxation in Germany

9. An assessment of the Riester pension plans

The introduction of the Riester pensions in 2001 offers a fascinating case study in the potential success and failure of tax-favored but voluntary retirement saving. Will workers overcome the temptations to procrastinate? How many will build up supplementary pensions? How much will they save?

At this point, only three years since their introduction, it is too early to tell. It took about about a decade to popularize a general subsidized dedicated savings program in Germany ("*Vermögenswirksame Leistungen*", directly deducted from payroll) which now enjoys almost universal participation. In the US, IRAs needed equally long to be accepted by a large share

of households, and participation is still biased towards the well-to-do. In this section, we look at the design and the incentives in order to understand who is likely to take up the newly created Riester pensions.

9.1 Depth of subsidies versus administrative restrictions

The first aspect is the depth of Riester incentives. As described in the preceding section, there are two kinds of incentives: the subsidies and tax exemptions during the contribution phase and tax-related advantages or disadvantages during the disbursement phase. The direct subsidies during the contribution phase are very deep for those who have relatively low income and those who have children. The reverse is the case for the tax-deductible special allowances, due to the progressive tax system. Here, households with higher incomes benefit more. This results in a U-shaped relation between subsidies and income, visible in figure 5 which shows the subsidy as a percentage of savings in form of the new supplementary pensions.²¹



Figure 5: Depth of subsidies to Riester pensions

Note: Direct subsidy/the tax advantage as a percentage of savings in form of the new supplementary pensions. *Source:* Deutsche Bundesbank (2002).

For lowest income households, the subsidy is almost as large as the contribution itself. Even for the well-to-do, subsidy rates are high around 40-50 percent. Given these deep subsidies, uptake is likely to be high. Figure 3, however, is only one part of the picture. The U-shaped

²¹ We use the word "subsidy" for both the direct subsidy and the tax-deductible special allowance.

curve will eventually be flattened out during the disbursement phase when pension benefits will be taxed. This flattening effect is due to the impact of progressive taxation. Taxation will not affect pensioners in the lower half of the income distribution because their pension income is below the still generous exemption for retired households. It will, however, considerably reduce the effective lifetime subsidy to households with incomes above average. The tax relief for Riester pensions, therefore, are heavily tilted towards those households who are less likely to invest in old-age provision, given the experience in other countries (notably the United Kingdom and the United States).

While the depth of the incentives makes Riester pensions rather attractive, and especially so for the less well-to-do, they are less flexible than other retirement investment products. One of the main complaints is that most of the capital has to be annuitized and can therefore not be used as collateral (e.g., for a mortgage) or be bequeathed. The argument lacks a certain logic since the very objective of the Riester pensions is to provide annuity income in order to fill the pension gap emerging from the reduced PAYG pillar. Nonetheless, the widely voiced argument is a clear indication that most workers have not yet realized that they will depend on Riester pensions for a reasonable retirement income.

The extensive certification requirements severely restrict private providers' scope to develop new private insurance products and lead to higher costs. Certain cost items can result in total costs of up to 20 percent, compared with around 10 percent for a normal capital sum life insurance policy.²² What is more, the certification rules merely serve to create a formal product standard without creating the transparency needed in order to compare different investment vehicles and the relative rates of return they offer. As a result, customers are often not in a position to make truly informed private investment decisions. The guarantee of the nominal value of contributions does ensure that, on retirement, at the very least the nominal capital saved is available as pension capital. However, there are set ways to provide indexation which is needed in order to ensure that the value of pension benefits paid out from the saved capital can be maintained over the long term. Riester benefits that are defined in nominal Euros will lose their value even at very modest rates of inflation given the long terms involved

²² Stiftung Warentest (2002).

9.2 Preliminary evidence on take-up rates

First survey results show that demand for Riester products is sluggish: only around 9 percent had actually taken out a policy by mid 2002; a further 16 percent planned to conclude a policy by the end of 2002. By the end of 2003, however, the take-up rate has increased to about 35 percent of all eligible workers.

This comes during a growing trend for workers to enroll in supplementary pension plans. Only around half of those planning to enroll in such plans are considering doing so in the framework of a Riester pension. The other half prefer other savings and insurance products, and/or occupational pensions.²³ Moreover, many households, especially in the higher income brackets, merely may restructure their existing pension plans in order to reap Riester subsidies. Börsch-Supan, Heiss und Winter (2004) provide evidence for such substitution. Slightly more than 50 percent in a representative sample of German workers who invested in Riester pensions claimed that they simply pushed funds from unsubsidized retirement savings into Riester pensions, while 37 percent answered that they created new savings.

Finally, there is consistent evidence that take-up rates of private retirement saving are much higher among those who are well informed about the pension system (know the contribution rate, know the pay-as-you-go mechanism, know that aging will put additional pressure on public pension systems), see Boeri, Börsch-Supan and Tabellini (2001, 2002a and b) and Börsch-Supan, Heiss und Winter (2004). Information and knowledge creation, therefore, seems to be an important incentive to boost retirement saving.

10. Summary and Conclusions

This paper provides a mixed bag of insights. First, the amount of retirement saving needed to fill the pension gaps created by pension reform all over Europe is substantial. Increasing the household saving rate by 4%, the benchmark in Germany, is a major feat.

Second, there is clear evidence tax relief shifts saving into those channels which enjoy the highest tax relief. Third, equally clear is the evidence on substitution between PAYG pensions and private retirement savings in the long run, as seen in the cross-national evidence. People eventually adjust their behavior to the generosity of public pension systems like in a system of communicating pipes. The current level of taxes appears to be uncorrelated with these ad-

²³ Leinert (2003).

justments. This does not necessarily imply that tax incentives boosting retirement savings are superfluous, since in the short-run, they might help in a transition to a new regime. The strong substitution effect suggests that information about the future level of PAYG benefits seems to be an important policy tool in order to instigate private retirement saving.

Fourths, the paper unfortunately adds little to the core question, whether tax relief creates additional new savings. This is in spite of the large institutional variation in Europe. While we do know that subsidies strongly increase saving in the specific form that is subsidized, possibly to the detriment of other saving forms, we do not really have firm evidence that savingrelated tax relief or similar subsidies increase total saving in Europe.

This does not make tax-relief a potentially wasteful instrument. Even if tax relief would only shift other saving to retirement saving, this may be a valuable mechanism if the government wants to make sure that the elderly will have a generous multipillar retirement income. That is, even if the government does not believe in the creation of new saving for macroeconomic purposes, it still may want to repress procrastination not only in form of consumption now, but also in the form of a larger house in the near future, and subsidize retirement consumption in the far future.

Finally, the German reforms in 2001 and 2004 have successfully installed a political process that will stabilize contribution rates to the PAYG public pension system, avoiding further harm to labor markets and economic growth. However, the jury is still out whether the resulting pension gap will be closed by tax-favored and subsidized retirement saving. Shifting to deferred taxation will cost the German government 6 billion Euros in lost revenues. So far, uptake rates are small. The German experience certainly teaches that tax relief can be made fruitless if it is combined with intransparent regulations restricting investment possibilities and withdrawal plans. In addition, information was poor. The German government tried to convince workers that the cuts in the PAYG pillar are relatively small – this certainly did not help the drive to start building up unprecedented second and third pillars.

To sum up, my reading of the evidence in Germany and Europe at large is that the best policy in order to boost retirement saving is a transparent description of the level of future PAYG pensions. Tax relief may serve well as an instrument to dampen liquidity problems especially among young families who have children and need to invest in homeownership. The crucial mechanism, however, is simply the substitution between declining PAYG pensions and retirement savings: Mind the pension gap!

References

- Abel, A.B., N.G. Mankiw, L.H. Summers, and R.J. Zeckhauser (1989), Assessing Dynamic Efficiency: Theory and Evidence, *Review of Economic Studies*, 56, 1-20.
- Alessie, R., and A. Kapteyn (2001): Household savings in the Netherlands. *Research in Economics*, 55, 61–82.
- Alessie, R., A. Kapteyn and F. Klijn (1997), Mandatory pensions and personal savings in the Netherlands, *De Economist*, 145, no. 3, pp. 291-324.
- Antolin, Pablo, Alain de Serres and Christine de la Maisonneuve (2004): Long-Term Budgetary Implications if Tax-Favoured Retirement Saving Plans, OECD Economics Department, Paris: OECD.
- Attanasio, O. and T. De Leire (1994), IRAS and household saving revisited: some new evidence, *NBER working* paper 4900.
- Attanasio O. and A. Brugiavini, (1997), L'effetto della Riforma Amato sul risparmio delle famiglie italiane, *Ricerche Quantitative per la Politica Economica-1995*, Banca d'Italia-CIDE (eds.), Roma
- Attanasio O., and J. Banks (1998), Trends in household saving don't justify tax incentives to boost saving, *Economic Policy*, October.
- Auerbach, A.J., and L.J. Kotlikoff (1983), An examination of empirical tests of social security and savings, in: E. Helpman et al. (eds.) Social Policy evaluation: an Economic Perspective, Academic Press, New York, 161-179
- Auerbach, A.J. and L.J. Kotlikoff (1987), *Dynamic Fiscal Analysis*, Cambridge University Press, Cambridge, England.
- Auerbach, A.J., W.G. Gale and P.R. Orszag (2003), Reassessing the Fiscal Gap: The Role of Tax-Deferred Saving, *Tax Notes* July 2003.
- Barro R. J. (1974), Are Government Bonds Net Wealth?, *Journal of Political Economy*, 82,5,1095-117
- Bernheim B.D. (1987a), The Economic Effects of Social Security: Toward a Reconciliation of Theory and Measurement, *Journal of Public Economics*, 33, 273-304
- Bernheim B.D. (1987b), Dissaving After Retirement: Testing the Pure Life Cycle Hypothesis, in Z. Bodie, J.Shoven and D.Wise, eds. *Issue in Pension Economics*, NBER, The University of Chicago Press, Chicago.
- Birg, H., and A. Börsch-Supan (1999), Für eine neue Aufgabenteilung zwischen gesetzlicher und privater Altersversorgung, GDV: Berlin.
- Boeri, T., A. Börsch-Supan, and G. Tabellini (2001): Would you like to Shrink the Welfare State? The Opinions of European Citizens, *Economic Policy*, Vol. 32.
- Boeri, T., A. Börsch-Supan, and G. Tabellini (2002a): Would you Like to Reform the Pension System? The Opinions of European Citizens, *American Economic Review*, May 2002.
- Boeri, T., A. Börsch-Supan, and G. Tabellini (2002b): How would you Like to Reform your Pension System? The Opinions of German and Italian Citizens, in: R. Brooks and A. Razin (eds.), *The Politics and Finance of Social Security Reform*, Kluwer, in press.
- Börsch-Supan, A. (1992): Saving and consumption patterns of the elderly: the German case. *Journal of Population Economics*, *5*, 289-303.

- Börsch-Supan, A. (1994a), Savings in Germany Part I: Incentives. In: Poterba, J. M. (ed.), *Public Policies and Household Saving*. Chicago: University of Chicago Press, 81-104.
- Börsch-Supan, A. (1994b), Savings in Germany Part II: Behavior. In: Poterba, J. M. (ed.), *International Comparisons of Household Saving*. Chicago: University of Chicago Press, 207-236.
- Börsch-Supan, A. (1999), Zur deutschen Diskussion eines Übergangs vom Umlage- zum Kapitaldeckungsverfahren in der Gesetzlichen Rentenversicherung. *Finanzarchiv*, Band 55, Heft 3, S. 400-428.
- Börsch-Supan, A. (2001), International Comparisons of Household Savings, *Research in Economics*, 55.
- Börsch-Supan, A., ed. (2003), *Life-Cycle Savings and Public Policy*, New York: Academic Press.
- Börsch-Supan, A. and A. Brugiavini, 2001, Savings: The Policy Debate in Europe, *Oxford Review of Economic Policy*, Vol. 17 No.1, 116-143.
- Börsch-Supan, A. und M. Lührmann (2000), Prinzipien der Renten- und Pensionsbesteuerung, Frankfurter Institut: Bad Homburg. English translation: Retirement benefit and pension taxation principles. In: O. Castellino and E. Fornero (eds.), Pension Policy in an Integrating Europe, Edward Elgar Publishing, 152-174.
- Börsch-Supan, A. und A. Lusardi, 2003, Saving: A Cross-National Perspective, In: *Life Cycle Savings and Public Policy*, Academic Press, New York, 1-31.
- Börsch-Supan, A. and M. Miegel, eds. (2001), *Pension Reform in Six Countries*, Springer: Berlin, Heidelberg, New York.
- Börsch-Supan, A. und A. Reil-Held (1998), Retirement Income: Level, Risk, and Substitution among Income Components. In: OECD (ed.), *Maintaining Prosperity in an Ageing Society*. Paris: OECD, AWP 3.7, 1-21.
- Börsch-Supan, A. und A. Reil-Held (2001): How much is Transfer and how much Insurance in a Pay-As-You-Go System? The German Case, *Scandinavian Journal of Economics* 103, S. 505-524.
- Börsch-Supan, A., and K. Stahl (1991a): Life-cycle savings and consumption constraints. *Journal of Population Economics*, 4, 233-255.
- Börsch-Supan, A. und K. (1991b): Do Savings Programs Dedicated to Homeownership Increase Personal Savings? An Analysis of the West German Bausparkassensystem. *Journal of Public Economics* 44, 265-297.
- Börsch-Supan, A. and C. B. Wilke, 2003, The German Public Pension System: How it Was, How it Will Be, Discussion Paper, MRRC, University of Michigan, and MEA, University of Mannheim.
- Börsch-Supan, A., A. Ludwig and J. Winter, 2003, Aging, pension reform, and capital flows: A multi-country simulation model, Discussion paper 28-2003, MEA, University of Mannheim.
- Börsch-Supan, A., A. Reil-Held and R. Schnabel (1999): Pension provision in Germany. In:P. Johnson (Hg.): *Pensioners' Income: International Comparisons*. Cambridge, MA and London: MIT-Press. Im Erscheinen.

- Börsch-Supan, A., A. Reil-Held, R. Rodepeter, R. Schnabel, and J. Winter (2001): The German Savings Puzzle, *Research in Economics*,
- Boskin, M.J. (2003): Deferred Taxes in the Public Finances, mimeo, Hoover Institution.
- Browning, M. and A. Lusardi (1996): Household saving: macro theories and micro facts. *Journal of Economic Literature*, 34, 1797-1855.
- Brugiavini, A. (1987), Empirical Evidence on Wealth Accumulation and the Effects of Pension Wealth: an Application to Italian Cross Section Data", 1987, *Financial Markets Group*, D.P. 20, LSE, London.
- Brugiavini, A. and M. Padula (2001): Too much for retirement? Saving in Italy. *Research in Economics*, 55, 39–60.
- Brunsbach, S. and O. Lang (1998), Steuervorteile und die Rendite des Lebensversicherungssparens, *Jahrbücher für Nationalökonomie und Statistik* 217, 185-213.
- Casey, B., H. Oxley, E. Whitehouse, P. Antolin, R. Duval and W. Leibfritz (2003): Policies for an Ageing Society: Recent Measures and Areas for Further Reform, OECD Economics Department Working Paper 369, Paris: OECD.
- Chiuri M.C and T. Jappelli (2000), Financial Markest Imperfections and Homeownership: an International Comparison, *CSEF working paper*, Salerno.
- Cigno, A., and F.C. Rosati (1996): Jointly Determined Saving and Fertility Behaviour: Theory, and Estimates for Germany, Italy, UK, and USA, *European Economic Review* 40, 1561-89.
- Deutsche Bundesbank (2002): Kapitalgedeckte Altersvorsorge und Finanzmärkte, Monatsbericht Juli, Frankfurt am Main.
- Deutsches Institut für Altersvorsorge (DIA, 1999): *Die Deutschen und Ihr Geld*. Köln: Deutsches Institut für Altersvorsorge.
- Dilnot, A.W. (1992): Taxation of Private Pensions: Costs and Consequences. In: OECD, *Private Pensions and Public Policy*, Paris: OECD.
- Disney R. and P. Johnson, (2001), *Pension systems and retirement incomes across OECD countries*, Edward Elgar, London.
- Disney, R. (1996), Can We Afford to Grow Older?, The MIT Press, Cambridge
- Disney, R., M. Mira D'Ercole and P. Scherer (1998), Resources during retirement, *OECD*, *Ageing Working paper*, AWP 4.3, Paris
- Elmendorf D.W and L.M. Sheiner, (2000), Should America Save for its Old Age? Fiscal Policy, Population Aging, and National Saving, *Journal of Economic Perspectives*, vol.14. no.3, 57-74
- European Commission, (1999) *Towards a single market for supplementary pensions*. Results of the consultation on the Green Paper on Supplementary pensions in the single market, COM(1999), 134.
- Euwals, R. (2000) 'Do Mandatory Pensions Decrease Household Savings: Evidence for the Netherlands.' *De Economist*, Vol. 148, pp 643-670.
- Eymann, A., and A. Börsch-Supan (2002): Household Portfolios in Germany. In: L. Guiso, M. Haliassos and T. Jappelli (eds.), *Household Portfolios*, Cambridge, Mass.: MIT-Press.

- Fall, M., C. Loisy, and G. Talon (2001): An empirical analysis of household savings in France 1984–1998. *Research in Economics*, 55, 155–172.
- Feldstein, M. (1974): Social Security, Induced Retirement and Aggregate Capital Accumulation, *Journal of Political Economy*, 82,5, 905-926.
- Feldstein, M. (1998). Privatizing Social Security. Chicago: The University of Chicago Press.
- Fenge, R. (1997): Effizienz der Alterssicherung. Frankfurt: Physica-Verlag.
- Gale W.J. and J.K. Scholz (1994) IRAs and household saving, American Economic Review, 84, 1233-1260
- Gruber, J., and D. Wise, eds. (1999): *Social Security and Retirement Around the World*. Chicago: The University of Chicago Press.
- Heller, P., 1989, Aging, Savings and the Sustainability of the Fiscal Burden in the G7countries: 1980-2025. Washington, D.C.: IMF.
- Hubbard R.G., (1986) Pension wealth and individual saving, some new evidence, *Journal of Money Credit and Banking*, 18, 167-178
- Jappelli T. (1995), Does social security wealth reduce the accumulation of private wealth? Evidence from Italian survey data, *Ricerche Economiche*, 49,1-31
- Jappelli T., and M. Pagano (1994), Saving Growth and Liquidity Constraints, *Quarterly Jour*nal of Economics, 109, 83-109.
- Jappelli, T., and F. Modigliani (1998). The age-saving profile and the life-cycle hypothesis, CSEF Working Paper No. 4, University of Salerno.
- Kapteyn, A., R. Alessie and A. Lusardi (1999), "Explaining the wealth holdings of different cohorts: productivity growth and social security", mimeo, *Tinbergen Institute*.
- Kim, S., (1992): Gesetzliche Rentenversicherung und Ersparnisbildung der privaten Haushalte in der Bundesrepublik Deutschland von 1962 bis 1988, Zeitschrift für die gesamte Versicherungswirtschaft 81, 555-.
- King M. and L. Dicks-Mireaux, (1982), Asset Holding and the Life Cycle, *Economic Journal*, 92
- Kohl, R. and P. O'Brien (1998), *The Macroeconomics of Ageing, Pensions and Savings: A Survey*, OECD Working Paper AWP1.1, Paris.
- Laboul, A. (1998), Private Pension Systems: Regulatory Policies, OECD Working Paper AWP2.2, Paris.
- Laibson D., A. Repetto and J. Tobacman (1998), Self-control and saving for retirement, *Brooking Papers in Economic Activity*, 1-98, pp. 91-196.
- Lang, O. (1998): Steueranreize und Geldanlagen im Lebenszyklus: empirische Analysen zu Spar- und Portfolioentscheidungen deutscher Privathaushalte. Baden-Baden: Nomos Verlag.
- Lang, O., K. H. Nöhrbaß, and K. Stahl, 1997, On Income Tax Avoidance: The Case of Germany. *Journal of Public Economics* 66, 327-347.
- Leinert, J. (2003): Die Riester-Rente: Wer hat sie, wer will sie: Vorausbewertung einer repräsentativen Umfrage zum Vorsorgeverhalten der 30- bis 50-Jährigen, Bertelsmann Stiftung Vorsorgestudien 14, Gütersloh

- Miles, D. (1999): Modeling the impact of demographic change upon the economy. *Economic Journal*, 109, 1–36.
- OECD (1988), Ageing Populations: The Social Policy Implications, Paris.
- OECD (1996), Future Global Capital Shortages: Real Threat or Pure Fiction? Paris: OECD, 103-141.
- Palacios, R. und R. Rocha (1998): The Hungarian Pension System in Transition, *Social Protection Discussion Paper Series* 9805, The World Bank, Washington, D.C., 1998.
- Poterba, J.M., S.F. Venti und D.A. Wise (1994): 401(k)-Plans and Tax-Deferred Saving, in: D.A.Wise (publisher): *Studies in the Economics of Aging*, Chicago: University of Chicago Press.
- Rossi N. and I. Visco (1994): Private Saving and the Government deficit in Italy, In: Ando A.,L. Guiso and I.Visco. (eds.), Saving and the Accumulation of Wealth, Cambridge: Cambridge University Press.
- Schnabel, R. (1999): *Ersparnis und Vermögen im Lebenszyklus in Westdeutschland*. Habilitation, Universität Mannheim.
- Sinn, H.W. (2000), Why to Fund and Why Not to Fund, mimeo, CES, Munich
- Skinner, J. und R.G. Hubbard (1996), Assessing the Effectiveness of Saving Incentives, Journal of Economic Perspectives 10(4), 73-90.
- Stiftung Warentest (2002): Riester-Rentenversicherungen: Die Lücke schließen, FINANZtest 9/2002
- Summers, L. (1989): Some Simple Economics of Mandated Benefits, American Economic Review, Vol. 79, No. 2, May 1989.
- Venti S. and D. Wise (1990), Have IRAs increased U.S. Savings? Evidence from consumer expenditure surveys, *Quarterly Journal of Economics*, 105, 661-98.
- Walliser, J. and J. Winter (1999): Tax incentives, bequest motives and the demand for life insurance: evidence from Germany. Discussion Paper No. 99-28, Sonderforschungsbereich 504, University of Mannheim.
- Wise, D.A. (2001), United States Support in Retirement: Where We Are and Where We Are Going, in: Börsch-Supan, A. and M. Miegel, eds. (2001), *Pension Reform in Six Countries*, Springer: Berlin, Heidelberg, New York.
- World Bank (1999): The Tax Treatment of Funded Pensions, *World Bank Pension Reform Primer* 2.3, Washington, D.C.
- Yoo, K.Y. and A. de Serres (2004): The Tax-Treatment of Private Pension Savings in OECD Countries and the Net Tax Cost per Unit of Contributions to Tax-Favoured Schemes, OECD Economics Department, Paris: OECD.

Nr.	Author	Title
07-19	Jijrgen Eichberger	Ambiguity and Social Interaction
07-17	David Kelsey Burkhard Schipper	Amonguly and Social Incraction
07-18	Jürgen Eichberger Willy Spanjers	Liquidity and Ambiguity: Banks or Asset Markets?
07-17	Patrick A. Müller Jana Janßen Dominique Jarzina	Applicantsí reactions to selection procedures ñ Prediction uncertainty as a moderator of the relationship between procedural fairness and organizational attractiveness
07-16	Patrick A. Müller Dagmar Stahlberg	The Role of Surprise in Hindsight Bias ñ A Metacognitive Model of Reduced and Reversed Hindsight Bias
07-15	Axel Börsch-Supan Anette Reil-Held Daniel Schunk	Das Sparverhalten deutscher Haushalte: Erste Erfahrungen mit der Riester-Rente
07-14	Axel Börsch-Supan Dirk Krüger Alexander Ludwig	Demographic Change, Relative Factor Prices, International Capital Flows, and their Differential Effects on the Welfare of Generations
07-13	Melanie Lührmann	Consumer Expenditures and Home Production at Retirement: New Evidence from Germany
07-12	Axel Börsch-Supan Anette Reil-Held Christina Wilke	Zur Sozialversicherungsfreiheit der Entgeltumwandlung
07-11	Alexander Ludwig Dirk Krüger	On the Consequences of Demographic Change for Rates of Returns to Capital, and the Distribution of Wealth and Welfare
07-10	Daniel Schunk	What Determines the Saving Behavior of German Households? An Examination of Saving Motives and Saving Decisions
07-09	Axel Börsch-Supan Anette Reil-Held Christina Wilke	How an Unfunded Pension System looks like Defined Benefits but works like Defined Contributions: The German Pension Reform

SONDERFORSCHUNGSBereich 504 WORKING PAPER SERIES